Author's response to reviews

Title: Inputs to Quality: Supervision, Management, and Community Involvement in Health Facilities in Egypt in 2004

Authors:

Emily J Cherlin (emily.cherlin@yale.edu)
Adel A Allam (aallamegy@netscape.net)
Erika L Linnander (erika.linnander@yale.edu)
Rex Wong (rex.wong@yale.edu)
Essam El-Toukhy (eeltoukhy@yahoo.com)
Heather Sipsma (heather.sipsma@yale.edu)
Harlan M Krumholz (harlan.krumholz@yale.edu)
Leslie A Curry (leslie.curry@yale.edu)
Elizabeth H Bradley (elizabeth.bradley@yale.edu)

Version: 2 Date: 1 February 2011

Author's response to reviews: see over
February 1, 2011

Jigisha Patel, MR CP, PhD
Series Editor
BMC Health Services Research

Dear Dr. Patel,

Enclosed is the revised manuscript # 2010577663509949 : “Inputs to Quality: Supervision, Management, and Community Involvement in Health Facilities in Egypt in 2004” for review by the BMC Health Services Research. The manuscript was submitted as an original research article. In recognition of BMC Health Services Research journal’s interest in articles that focus on the management of health services. We made one more change in the Methods section of the Abstract in the following line “We conducted a cross-sectional analysis of data from xxx facilities surveyed with the Egyptian Provider Services Assessment (ESPA) survey in 2004”. We replaced the xxx with the number of facilities (559).

This paper is not under review by any other journal and all co-authors have approved this manuscript for submission. Please contact me at Elizabeth.bradley@yale.edu if you have any questions about the paper or its submission. Thank you for advance for your review.

Sincerely,

Elizabeth H. Bradley, PhD
Professor of Public Health
Director, Health Management Program
and Yale Global Health Initiative